

IWIM - Institut für Weltwirtschaft und Internationales Management



IWIM - Institute for World Economics and International Management

INCOME DISTRIBUTION AND ACCESSIBILTY TO PRIMARY AND SECONDARY SCHOOLS IN NIGERIA

Reuben Adeolu Alabi

Institute for World Economics and International Management (IWIM)

University of Bremen, Bremen, Germany

Alexander von Humboldt Research Fellow

E-mail: <u>bayobimb@yahoo.com</u> Phone Number: +491623846199

Berichte aus dem Weltwirtschaftlichen Colloquium der Universität Bremen

Nr. 114

Hrsg. von

Andreas Knorr, Alfons Lemper, Axel Sell, Karl Wohlmuth



INCOME DISTRIBUTION AND ACCESSIBILTY TO PRIMARY AND SECONDARY SCHOOLS IN NIGERIA

Reuben Adeolu Alabi¹

Andreas Knorr, Alfons Lemper, Axel Sell, Karl Wohlmuth (Hrsg.):

Berichte aus dem Weltwirtschaftlichen Colloquium der Universität Bremen, Nr. 113, Juli 2008 ISSN 0948-3829

Bezug: IWIM - Institut für Weltwirtschaft

und Internationales Management

Universität Bremen

Fachbereich Wirtschaftswissenschaft

Postfach 33 04 40 D- 28334 Bremen

Telefon: 04 21 / 2 18 - 34 29 Telefax: 04 21 / 2 18 - 45 50 E-mail: iwim@uni-bremen.de http://www.iwim.uni-bremen.de

¹ This paper would not have seen the light of the day without the supervision and support of my mentor and supervisor (Prof . Dr. Karl Wohlmuth). The support of other staff of IWIM, University of Bremen, such as Prof. Dr. Sell and Mister Osmund Uzor is also acknowledged. Corinna Hartmann provided for the research the necessary technical support. I am grateful to all of them. The usual disclaimer applies.

Zusammenfassung

Die Studie untersucht die Beziehungen zwischen Einkommen, Einkommensun

gleichheit und dem Zugang zur Grund- und Oberschule in Nigeria. Sie stellt heraus,

dass es einen ungleichen Zugang zur Grundausbildung zwischen armen und reichen

Menschen in Nigeria gibt. Das Haushaltseinkommen ist eine wichtige Bestimmungs-

größe für den Zugang zur Grundausbildung in Nigeria. Dabei kann eine Zunahme im

Zugang zur Grundausbildung das Einkommen in Nigeria schneller umverteilen als

eine Zunahme im Haushaltseinkommen. Die Wirkung der Einkommensumverteilung

auf den Zugang zur Oberschule ist dabei größer als auf den Zugang zur Grundschule.

Diese Studie schlussfolgert, dass eine Politik mit dem Ziel eines Ausgleiches beim

Zugang zur Grund- und Oberschule helfen könnte, die Einkommensungleichheit in

Nigeria zu reduzieren. Sie empfiehlt neben anderen Dingen Richtlinien und Pro-

gramme, die den Zugang zur Grundausbildung in Nigeria erhöhen.

Abstract

This study estimated the relationship between income, income inequality and

accessibility to primary and secondary schools in Nigeria. It establishes that there is

an unequal access to basic education between the poor and non poor in Nigeria.

Household income is an important determinant of access to basic education in Nige-

ria. Increase in access to basic education can redistribute income in Nigeria faster than

increase in household income. The income redistribution effect of accessibility to sec-

ondary school is greater than primary school. This study concludes that a policy aim-

ing at equalisation of access to primary and secondary school education might help in

reducing income inequality in Nigeria. It recommends among other things, policies

and programmes that will increase access to basic education in Nigeria.

Keywords: Accessibility, Primary, Secondary, School, Nigeria

Stichwörter: Zugang, Grund, Ober, Schule, Nigeria

JEL-Classification: I20, I21

TABLE OF CONTENTS

L	IST OF TABLES AND FIGURES	II
L	IST OF ABBREVIATIONS	II
1	INTRODUCTION	1-2
1	INTRODUCTION	1-2
2	HOUSEHOLD INCOME AND SCHOOLING	2-4
3	HOUSEHOLD INCOME AND SCHOOLING INVESTMENT	4-8
4	EDUCATION AND INCOME DISTRIBUTION	8-11
5	RECENT EVIDENCE ON INCOME DISTRIBUTION AND	11-14
	EDUCATION	
6	RESEARCH METHODOLOGY	14-18
7	RESULTS AND DISCUSSIONS OF DESCRIPTIVE RESULTS	19-27
8	RESULTS AND DISCUSSIONS OF ECONOMETRIC RESULTS	27-30
9	CONCLUSIONS AND RECOMMENDATIONS	31-32
	REFERENCES	33-40

LIST OF TABLES AND FIGURES

Table 1:	Some Past Studies on Income and Schooling	5
Table 1.	Some I ast Studies on meome and senooning	3
Table 2:	Income Differences According to Educational Level in Selected African Coun-	12
tries		
Table 3:	Accessibility to Primary and Secondary Schools on Regional Basis in Nigeria (%)	20
Table 4:	Accessibility to Basic education in Urban and Rural Areas in Nigeria (%)	22
Table 5:	Accessibility to primary and secondary education and Income distribution in Nigeria	23
Table 6:	Accessibility to primary and secondary education among the poor in rural and urban areas in Nigeria	24
Table 7:	Effect of Household income on accessibility to Basic Education in Nigeria	25
Table 8:	Effect of Accessibility to Basic Education on Income Inequality in Nigeria	26
Table 9:	Effect of Mean Household Income on Income Inequality in Nigeria	27
Figure 1:	Map of Nigeria	14

LIST OF ABBREVIATIONS

EA ENUMERATION AREA EFA EDUCATION FOR ALL

CWIQ CORE WELFARE INDICATOR
FCT FEDERAL CAPITAL TERRITORY

HU HOUSING UNIT

NAR NET ENROLMENT RATE

NBS NATIONAL BUREAU OF STATISTICS

UNECA UNITED NATION ECONOMIC COMMISION FOR AFRICA

UNICEF UNITED NATIONS CHILDREN'S FUND

UNESCO UNITED NATIONS EDUCATIONAL, SCIENTIFIC, AND

CULTURAL ORGANISATION

1 INTRODUCTION

The importance of education to human beings cannot be over emphasized. Education is a human right that should be accorded to all human beings solely by reason of being human (Igbuzor, 2008). There are a lot of international human rights instruments that provide for education as a fundamental human right. These include the Universal Declaration of Human Rights (1948), the International Covenant on Economic, Social and Cultural Rights (1966) and the African Charter on Human and Peoples' Rights (1981). The relationship between education and development is well established such that education is a key index of development. It has been documented that schooling improves productivity, health and reduces negative features of life such as child labour as well as bringing about empowerment (EFA Global Monitoring Report, 2002). Interestingly, the role of education has also been recognised in the discourse on the causation of civil wars. Some empirical evidence shows that civil wars are concentrated in countries with little education and importantly a country with higher percentage of its youth in schools reduces considerably its risk of conflict (Collier, 2000). This is why there has been a lot of emphasis particularly in recent times for all citizens of the world to have access to basic education.

The importance and linkage of education to the development of any society is well known. It is in recognition of this importance that the international community and governments all over the world have made commitments for citizens to have access to education. Meanwhile, it has been documented that across the globe, there are inequalities in educational access and achievement as well as high levels of absolute educational deprivation of both children and adults (Subrahmanian, 2002). Despite the apparent recognition of the positive role of education in human development, improving access to education has been elusive across the globe particularly in the developing countries and specifically Sub-Sahara Africa. Equally, the 'right' to education that has recently been invoked in the lexicon of many development actors concerned with improving access to education is far from being realised and it remains a rhetoric rather than tangible reality(Deng, 2003). In order to confront this challenge, the rights based approach, which emphasizes the participation of citizens, has been advocated. Meanwhile, the Declaration of the World Conference on Education for All (WCEFA) which was made in Jomtien, Thailand in 1990 stated clearly in Article 1 that every person – child, Youth and Adult – shall be able to benefit from educational opportunities designed to meet their basic needs. This declaration was reaffirmed at the World Summit for Children also held in 1990, which stated that all children should have access to basic education by the year 2000. Similarly, the Millennium Developments Goals (MDGs) adopted in September 2000 at the United Nations Millennium Declaration has two of the eight goals devoted to education. They are goal 2 (to achieve universal primary education) and goal 3 (to promote gender equality and empower women).

Over the years, Nigeria has expressed a commitment to education, in the belief that overcoming illiteracy and ignorance will form a basis for accelerated national development. However, regardless of the incontrovertible evidence that education is crucial to the development of the community and the nation, there remain inequalities in access to education in Nigeria. This inaccessibility to basic education can be predicated on low income and high income inequality, which is growing in Nigeria. Location and regional disparities may also be pronounced in distribution of education resources in Nigeria. However, since generalization does not make sense, therefore is essential that the role of the income, income inequality, location and regional effects in explaining accessibility to education be investigated. Hence, this study will examine the effects of these variables in accessibility to primary and secondary education in Nigeria. The rest of the paper is divided in 9 sections. Following introduction is information on household income and schooling in section 2, section 3 examines household income and schooling investment, section 4 deals with education and income distribution, section 5 reviews recent evidence on relationship between income distribution and education, section 6 presents the research methodology, section 7 concentrates on discussion of descriptive results, section 8 presents and discusses econometric results, while section 9 concludes the paper.

2 HOUSEHOLD INCOME AND SCHOOLING

Schooling is widely seen as critical to the development process and poverty alleviation. Recent studies confirm that schooling is particularly important when complex new technologies and market options become available (Rosenzweig,1995). Recently, many countries have done considerable macroeconomic stabilization and market liberalization programs. The returns to schooling will probably increase following such programs. Therefore decisions about who is schooled now are likely to be critical in determining a country's future economic growth and distribution of income. A

rising concern for many in developing countries has been the possibility of greater inequality and reduced intergenerational social mobility under these economic reforms. Part of this concern is that family "dynasties" will be reinforced if children from higher-income households are more likely to receive more and better schooling, and thus reap greater gains from schooling in the future than children from lowerincome households. Two different societies with the same income distribution at a point in time may be viewed as having different levels of social welfare if they have different degrees of social mobility. For example, Friedman (1962) argues that a given extent of income inequality that arises in a rigid system in which each family stays in the same position each period may be a cause for more concern than the same degree of income inequality that arises in a fluid system because of the great mobility and dynamic change associated with equality of opportunity. Because of the concern that schooling could perpetuate social immobility and inequality, the recent policy-related literature has considered targeting public school resources toward children from poorer families (van de Walle and Nead 1995 provide examples and references). The concerns in developing countries have been about whether family dynasties are becoming more powerful and whether schooling is targeted toward children from poorer households or if it is instead reinforcing the advantages of children from better-off households. Educational reforms have exacerbated these concerns (see World Bank 1996). The reforms are intended to make schools more efficient, but some of their components (such as the introduction of user charges) may affect children differently depending on their household income.

Researchers have conducted numerous studies of associations between indicators of household income and schooling for other countries. Behrman and Knowles (1997) reviewed 42 studies, covering 21 countries. The studies that are related to African countries are summarized in Table 1. Of the cases for which they can estimate income elasticities, the median elasticity is 0.20. This number suggests that children from higher-income households do better in school than children from poorer households, although the magnitude of the effect is small. The estimates tend to be higher for samples with poorer households, and a number of the studies find small inverse associations between schooling and income. The largest elasticity estimates, those higher than 0.20-are for low-income regions (low-income during the period of the survey): Cote d'Ivoire and Ghana. But these are the only cases in which the estimates exceed 0.20.

Table 1: Some Past Studies on Income Elasticity of Schooling

Country /Year	Schooling Indicator	Income Elasticity	Source
Cote d'Ivoire (1985-87)	Completed years/current enrolment	0.19	Montgomery and Kouame (1993)
Cote d'Ivoire (1985-87)	School attainment	0.14 to 0.42	Tansel (1997)
Egypt (1980)	Ever attended school/currently attending	-	Cochrane, Mehra and Osheba (1986)
Ghana(1988-89)	Grade attainment, reading, dropping out age	-	Glewwe and Jacoby (1994, 1995)
Ghana(1987)	Ever attended school/school attainment	-	Lavy (1996)
Ghana (1987-89)	School attainment	0.18 to 0.56	Tansel (1997)
Kenya(1994)	Enrolment, Student- teacher ratio	0.04	Deolalikar (1997)
Mali(1981-1982)	Enrolment	0.38	Birdsall and Orivel (1996)
South Africa(1993)	Years of schooling	-0.01 to 0.10	Jacoby (1994)

Source: Adapted from Behrman and Knowles (1999)

3 HOUSEHOLD INCOME AND SCHOOLING INVESTMENT

This discussion is based on how schooling investment might be associated with household income. This discussion points to a number of possible reasons, as well as to the difficulty of disentangling association and causality from cross-sectional data and of determining whether such associations may reflect underlying inefficiencies. If there were no unobserved differences between low- and high-income households, if schooling were purely an investment (with no current consumption as-

pects), if markets worked perfectly, and if the same prices prevailed in all markets, there would be no differences in schooling investments associated with income once controlling for any observed differences in household characteristics. Therefore it is useful to determine why there might be associations between household income and investments in schooling. The general reasons are that household income is proxying for correlated unobserved determinants of child schooling, such as innate ability, preferences, and family connections; household income is proxying for price variations in school inputs; and household income is playing a causal role in the presence of imperfect markets. In addition to the investment aspect of schooling, spending time in school may be a current consumption activity that is associated with household income. Schooling may also affect future consumption (for example, by enriching reading as an adult), but because these effects are obtained in the future, current schooling for such purposes is an investment. If the current consumption of schooling has aspects that are normal goods, ceteris paribus, more household income leads to more schooling for that reason alone.

The relationship between schooling as an investment and household income is multifaceted and more complicated than the relationship between schooling as current consumption and household income. Becker's (1967) lecture on the determinants of human capital investments is a useful starting point for thinking in more detail about possible associations between parental household income and schooling investments. Within this framework schooling investments are made until the private marginal benefits of the investment equal its private marginal costs. If all markets function perfectly, there are no government interventions, and schooling is only an investment, then everyone invests in schooling until the expected rate of return from schooling equals the expected rate of return on alternative investments, regardless of household income. In this case there are no or very few channels through which income may be associated with schooling. But given the range of real-world market imperfections and government interventions, there are many reasons why household income may be associated with schooling, even if schooling is purely an investment. To illustrate, consider what would happen in the presence of market imperfections. There are several explanations, originating in both policy and market failures (as well as reasons that would persist with perfect markets), why household income may be related to the marginal private benefits and costs of schooling investments and thus to schooling investments themselves. Current consumption effects could also generate associations between income and schooling (with the sign depending on the nature of the consumption effects). Some of these reasons reflect inefficiencies, such as those due to imperfect credit and information markets. Others reflect differing abilities that complement human capital investments or differing prices that are related to household income in different areas given positive transportation costs. Some reflect causal effects of income, such as current consumption demands. And some reflect associations with other variables, such as abilities that are correlated with income and transferred in part inter-generationally. With cross-sectional data of the types that are usually available, the relevance of many of these possibilities and the effect of causality compared with association cannot be sorted out conclusively. Why might marginal private benefits of schooling be associated with household income in the presence of government policies or market imperfections? There are several reasons. First, public policies may affect households with different incomes differently. Policies may favor higher-income households by offering them higher-quality (or more accessible) schooling in response to their greater economic and political power or because prices of some important school inputs may be lower in areas where incomes are higher (for example, teachers may prefer to live and teach in high-income areas and be willing to do so at lower salaries than they would require in low-income areas). However, policies may favor poorer households if programs are designed to reduce inequality or to alleviate poverty by allocating better schooling to poorer households or if prices of some school inputs are lower in low-income areas. Second, households may invest in children's education at home directly through tutoring or indirectly through improvements in their health and nutrition. If markets for these investments (or for financing these investments) are imperfect and the costs are lower for wealthier households, the marginal private benefits of schooling will be higher for wealthier households. For instance, the cost of helping with homework may be less for more-schooled parents than for less-schooled parents, and parental schooling is likely to be positively correlated with household income. Third, children's genetic endowments, for which there are no perfect markets (marriage markets probably serve indirectly as imperfect markets for such endowments), may interact with schooling investments and be correlated with parental endowments that, in turn, are correlated with household income. These relationships arise because such endowments affect income directly and indirectly through parents' human capital stock, including their education. Behrman et al (1996) present evidence, using data on twins, that schooling investments respond positively

to children's genetic endowments in the United States. Behrman and Taubman (1989) present estimates that variations in such endowments are consistent with most of the variance in child schooling for young adults in the United States. The enormous literature on the associations between adults' schooling and their household earnings is surveyed in Psacharopoulos (1994) and Rosenzweig (1995). Fourth, households may make complementary investments in searching for a job and have contacts that affect their children's job search after completing schooling. If markets for financing such investments are imperfect and the costs are lower for higher-income households, in part because of more attractive possibilities for working in family enterprises and better connections for other employment opportunities, the marginal private benefits would again be higher for such households. Fifth, higher-income households may have better information (in part because of better family enterprise options and better connections), given imperfect markets for information. As a result, they face less uncertainty about schooling investment decisions and, assuming constant risk aversion, therefore have higher expected marginal private benefits than poorer households. Sixth, higher-income households may have lower risk aversion. Therefore in the presence of imperfect insurance markets or insurance with positive private costs, their private incentives would be to invest more in schooling than otherwise identical lowerincome households. And lastly, higher-income households may be better able to deal with stochastic events. For example, through their connections (perhaps facilitated by income transfers, including bribes), they may be better able to offset their children's bad performance on admissions examinations than poorer households can. They therefore have private incentives to invest more in schooling than otherwise identical lower income households. The first possibility (involving public policies) relates to endogenous policy choices, which, depending on the mechanism, could favor either higheror lower- income households (see, for example, Rosenzweig and Wolpin, 1986). In the other six cases, higher-income households have private incentives to invest more in the schooling of otherwise equal children because they cope better with market imperfections, or higher-income households have unobserved characteristics that increase schooling investments and are associated with household income. Why might marginal private costs for human capital investments be associated with household income in the presence of market imperfections? Because of capital market imperfections, particularly for human capital investments (in part because human capital is not recognized as collateral), the marginal private costs for such investments are particularly high for individuals from poorer families who cannot as easily finance these investments themselves.

4 EDUCATION AND INCOME DISTRIBUTION

Becker and Chiswick (1966) demonstrate (in the US) that income inequality is positively correlated with schooling inequality and negatively correlated with the average level of schooling as indicated in Zilcha and Viaene (2003). Later, based on cross-section data from nine countries, Chiswick (1971) shows that earnings inequality increases with educational inequality. Later studies, based on larger sample of countries, support this result showing as well that higher level of schooling reduces income inequality (Chenery and Syrquin, 1975). Though human capital formation is a complex process, economic models have assumed some particular mechanisms describing it. Due to tractability reasons, these processes concentrate on very few parameters (Hanushek, 2002). According to Zilcha and Viaene (2003), the production function framework for human capital exhibits two important properties. First, individuals from below-average human capital families have a greater return to investment in public schooling than those from above-average families. In addition, the effort, and therefore cost, of acquiring human capital for the younger generation is smaller for societies endowed with relatively higher levels of human capital (Fischer and Serra, 1996). Second, the importance of parental human capital in forming the human capital of a child has been established (see, e.g., Hanushek, 1986). For example, Glaeser (1994) divides the education's positive effects on economic growth into parts, and concludes that children in families with educated parents obtain a better education than children without support. Also, Burnhill et al. (1990) find that parental education influences entry to higher education in Scotland over and above the influence of parental social class. Lee and Barro (2001) find that family characteristics, such as income and education of parents, enhance student's performance. A reason that is put forward is that parental education elicits more parental involvement (including related private investment) at home.

Gundlach et al. (2001) find that a higher stock of human capital increases the income of the poor, not only through its effect on average income, but also through its effect on the distribution of income. They interpret their findings as suggesting that effective education policies would be a first-best poverty reduction strategy. Ram

(1989) reviews several theoretical frameworks linking the level of schooling and its dispersion with income inequality, such as human capital or dual-economy type models. He finds that these models do not generate any clear theoretical hypotheses about the effect of education on income inequality or absolute poverty. For instance, traditional human capital models of earnings provide two opposing insights with regard to the relationships between education and income distribution. First, holding other things equal, these models imply a partial positive relation between the mean level of schooling and earnings inequality, such that if the mean level of schooling rises, wage of educated workers group relative to wages earned by non-educated worker will rise. But these models also feature a partial positive relation between schooling inequality and earning inequality in that a more equal distribution of schooling leads to a more equal distribution of earnings.

Knight and Sabot (1983) show these effects in a dual economy version of the human capital model. Educational expansion has gain two different effects on the distribution of earnings and thus on overall income inequality as it raised the supply of educated labour. On the one hand, the composition effect (or Kuznets effect) increases the relative size of the group with higher education (and higher earnings) and thus tend to increase inequality. On the other hand the wage compression effect resulting from the relatively greater supply of educated labour reduces inequality, which effect dominates is again unclear and will ultimately depend on the country's level of development, the relative size of the different educational groups, the degree of substitutability between workers with different levels of education, and the wider social, political and economic aspects that affect the structure of relative wages for different educational groups and the demand for labour.

To the extent that formal schooling is a significant component of human capital investment, the recent endogenous growth literature might provide a more conclusive theoretical framework regarding the relationship between educational expansion and income distribution. Tamura (1991) explains income convergence in the developed world by an endogenous growth model with human capital spillovers and heterogeneous agents. In his model, human capital convergence results in income convergence. Human capital convergence can be induced by educational expansion and the promotion of research activity, and because for a given stock of existing knowledge, agents with below average human capital have a higher rate of return to human capital investment. With a more explicit focus on the formal schooling component of

human capital investment, Glomm and Ravikumar (1992) construct an overlapping generations model with heterogeneous agents that provides similar results. The human capital possessed by each individual agent is a function of the parents stock of human capital, the level of schooling acquired, and the quality of education provided, which is modelled as an increasing function of tax revenue and determined endogenously by majority-voting. Further, they assume that the learning technology exhibits at least constant returns to the quality of schools and the parents stock of human capital. While they are mainly interested in comparing the effects of public and private investment in human capital on growth and the distribution of income, they also show that income inequality unambiguously declines over time in an economy with a public education sector where the quality of schooling is homogeneous. Since the growth rate of any agent's income is inversely related to his initial level, income convergence results in their model. By contrast, the endogenous growth model suggested by Luca (1988) does not predict income convergence. In this model, the human capitals supposed to generate internal and external effects. Where the latter means that the average level of education also contributes to the productivity of all other factors of production. However, the recent paper by Gunther (2002) shows that public education, its finances and the way it is undertaken are important determinants of income inequality and growth. In the model human capital simultaneously determines growth and income inequality. In this framework the paper identifies two redistribution mechanisms. On the one hand redistribution occurs by means of direct fiscal redistribution from the well-off to the not so well-off. On the other hand there is redistribution through taxes used for expenditure on public education, which redistributes income by changing the relative wages.

The centrality of education in poverty-reduction policies stems from the belief that education is a powerful equalizer. Human capital is supposed to generate internal and external effects, where the latter means that the average level of education also contributes to the productivity of all other factors of production. Gundlach et al (2001) estimates suggest that a 10% increase in the stock of quality adjusted human capital per worker would increase the average income of the poor by an additional 3.2%. Education seems to improve the income distribution and thus may allow the poor to benefit from growth to the great extent. Accordingly, a focus of economic policies on education in order to reduce poverty and to speed up development appears to be justified (Gundlach et al, 2001). The empirical findings indicate that improving the quality

of education rather than merely expanding access to education should play a crucial role in development strategies. According to Okojie (2002), the important strategy for reducing female poverty is great educational human capital investment in women. This will increase their access to higher paying and higher status job, thereby reducing household as well as female poverty, in addition to other non-market benefit of education. It is widely recognized today that human capital, in particular that acquired through schooling, is a key factor for development. The link is clearly established at the micro-economic level. Individuals with more education receive on average more income (Schultz, 1999). This implies that a more equalitarian distribution of education may constitute an efficient means of reducing irregularity of income distribution (Glomm and Ravikurmar, 1992). Likewise, in the Millennium Development Goals, education is seen as a powerful instrument not only 'for reducing poverty and inequality but also for improving health and social well-being, laying the basis for sustained economic growth, and being essential for building democratic societies and dynamic, globally competitive economies' (United Nations, 2000).

According to Gordon and Catherine (2001), there are several processes that reinforce the effect of education on incomes. Education increases skill levels, which are required for some activities, or contribute to increased productivity, or may be an employment rationing device. Education can set in train processes that increase confidence, establish useful networks or contribute to productive investment (exposure outside the home village, migration, using improved earnings to educate other family members or invest in rural enterprise). Education tends to be closely correlated with other variables that also improve access to higher income employment (pre-existing wealth, useful social networks and confidence). Better educated individuals are more likely to migrate to take up employment opportunities in other areas, as they have greater chances of success than their less-educated or uneducated counterparts. Islam (1997) argues that primary education enhances the productivity of the workforce, whilst secondary education stimulates entrepreneurial activity.

5 RECENT EVIDENCE ON INCOME DISRIBUTION AND EDUCATION

Education is a process through which mankind transmit experience, new findings, and values accumulated over time with the aim of enabling individuals and so-

cieties to make all rounded participation in the development process. In this regard, education plays a key role in enhancing economic progress, improving individual welfare and social development (Hannum and Buchmann, 2005). Available evidence shows that there are several channels through which such effects may arise. For instance education raises labor productivity, increases technological innovation and adaptation, contributes to better health (World Bank, 1993) and gives greater ability to deal with shocks. As a result, education is to date a basic ingredient for creating a competitive and knowledge based economy (Abebaw et al, 2007). Cogneau and Mesple-Somps (2008) have demonstrated that Ghana and Guinea have a lower income inequality compared with Ivory Coast, Ghana, Madagascar and Uganda because of the combination of widespread secondary schooling, low returns to education and low income dualism against agriculture. Nevertheless, it displays marked regional inequality insofar as being born in the Northern part of this country produces a significant restriction of income opportunities. Table 2 indicates that the mean income difference between non farmers with low education and those with high education is 96.

Table 2: Income Differences According to the Level of Education in Selected African Countries ²

Group	Ivory Coast	Ghana	Guinea	Madagascar	Uganda	Mean
Non Farmer with Low education	158	138	236	173	200	181
Non Farmer with High education	321	199	268	314	285	277
Difference	163	61	32	141	85	96

Source: Computed from Cogneau and Mesple Somps (2008).

In relating education with income distribution, Dercon et al. (2005) indicated that in Ethiopia, public sector workers were generally the best educated. They showed that far more secondary and university graduates were employed there than any other. They also showed that, while wages were increasing by education, the high variance in wages meant that only at the highest level of education were wages demonstrably

² Income is in Purchasing Power Parity terms.

(statistically) different. These findings suggest low or zero returns to education at lower levels of education and higher returns at higher levels. They revealed that wage dispersion between levels of education is far larger in the private sector. In Ethiopia, education seems to have had a substantial effect on allocation between sectors. For example, having completed primary education increases the probability of entering the public sector by 3 to 6 percent relative to having no education. Education is clearly linked with an intention to work in Ethiopia: having at least secondary education strongly reduced the probability of being out of the labour force. Similar evidence is obtained by Cockburn and Dostie (2007) who finds that child time allocation decision among schooling, and work in rural Ethiopia is strongly determined by a combination of household income and wealth, family composition and asset ownership.

Recently Abebaw et al. (2007) demonstrated that the major demand side factors determining schooling progress in rural Ethiopia include poverty, parental education, land and non-land asset ownership, village fixed effects and a child's demographic characteristics. On the supply side, differences in (accessibility) availability of primary and junior schools in the village significantly explain variation in children's primary education achievement. Previous other recent studies (Gitter and Barham, 2007; Gitter and Barham, 2007; Lire, 2005) find several explanations for the inadequate schooling and educational attainments of children particularly in developing countries of Africa, Asia, and Latin America. A common thread running through these studies is that child schooling experience in rural areas is related negatively with household poverty, and child age. Regarding gender effect, most studies just mentioned above find that girls are more likely to get less schooling than boys and that parental education has a positive and significant influence on enrolment and level of educational attainment.

Using a panel data from Tanzania, Burke and Beegle (2004), show that child school attendance is determined by a host of factors including household, child, and community characteristics. But they also noted that there are important gender differences in the factors influencing child schooling attendance and participation. In their analysis of the determinants of primary enrolment in Kenya, Bedi et al. (2004) have shown the key role played by child age, parental education, household wealth and school inputs on parents' decision to enrol their children.

The recent literature addressing the African growth problems stresses,

amongst other things, that the suboptimal level of public investments in education ha mpers growth and a more equitable distribution of income (Frankema and Bolt, 2006). The ruling elites face little incentives to direct public resources to education, since education can function as an important vehicle for people to organise themselves and become politically involved. Underinvestment in public education may be part of an intentional strategy to repress political opposition (Frankema and Bolt, 2006). Studies investigating the development of education in Africa indeed report that attainment levels are comparatively low, even when controlled for GDP per capita. It is also reported that the extent of educational inequality in Africa is extraordinary high. Sahn and Stifel (2004) further point out that the extent of educational inequality is significantly greater and that attainment levels are significantly lower in rural areas than in urban areas. Lloyd and Hewett (2004) find that African countries have the lowest primary completion rates of any region in the world. Regarding the levels of primary school completion they argue that "the poor are the least likely to send their children to school and their children, when enrolled, are most likely to perform poorly and drop out" Lloyd and Hewett (2004:14). Lloyd and Hewett emphasize that there is a strong interdependence between income levels and educational performance: there is a reciprocal causal relation between educational inequality and income inequality. Frankema and Bolt (2006) agree there is a negative relationship between educational attainment and educational inequality in Africa, but when they distinguish between attainment and distribution they find that attainment levels are much more important in explaining income inequality. Apparently the low levels of attainment in Africa influence income inequality mainly because of the barriers it poses to political and social reforms inhibiting the redistribution of income and resources. The impact of educational inequality on the distribution of direct income-generating capacities (i.e. human capital) only plays a modest role.

6 RESEARCH METHODOLOGY

6.1 Data Source and Collection

The data for this study is generated from Nigeria. Nigeria lies between 4^016^1 and 13^053^1 North Latitude and between 2^040^1 and 14^041^1 East Longitude. It is located in the West Africa bordered on the West by the Republic of Benin, on the north by the Republic of Niger and on the east by the Republic of Cameroon. To the South, Nige-

ria is bordered by approximately 800 kilometres of the Atlantic Ocean, stretching from Badagry in the West to the Rio del Rey in the east. The country also occupies a land area of 923,768 kilometres and the vegetation ranges from mangrove forest on the coast to desert in the far north. The map of Nigeria is presented in Figure 1.

Administration-wise, Nigeria consists of 36 states and a Federal Capital Territory. Each state is further divided into Local Government Areas (LGAs). These are 774 LGAs in the country. Nigeria returned into democratic rule in May 1999 under presidential system of government at federal, state and local government area levels. The federal government comprises of an Executive arm, a bicameral legislative arm and the judiciary. Each state has her own executive arm and house of assembly while each local government has a chairman and a council. The total population of Nigeria according to 2005 census was about 140 million.

The relevant data that are related to income and education are extracted from database obtained from the Nigeria Bureau of Statistics (NBS) Core Welfare Indicator Questionnaire (CWIQ) Survey of 2006. The Surveys ware conducted with assistance from European Union, World Bank, Department for International Development and United Nations Development Programme to ensure good quality of the data generation. The surveys had a national coverage, that is, all the 36 states of the Federation including the Federal Capital Territory of Abuja were covered. The sample design for the survey was a two stage stratified sample design. The first stage was the division of each state into clusters called Enumeration Areas (EA), while the second stage was the division of enumeration areas into housing units. One hundred and twenty (120) EAs were created for each state and 60 EAs for the Federal Capital Territory for the twelve months survey duration. Ten EAs for each state and five EAs for the FCT were covered per month (The survey was conducted through the twelve months period). The Core Welfare Indicator Questionnaire Survey (CWIQ) is designed to collect household data useful in quantitatively and quantitatively profiling the well-being of the population. The 2006 Nigerian CWIQ was a nationwide sample survey conducted to produce welfare indicators for the population at national and sub-national levels, particularly Zones, States and Senatorial Districts. The Survey complements 2004 Nigerian Living Standards Survey (NLSS) by NBS which profiled poverty in the country. Both surveys succinctly provide information for evidence-based policy actions as well as monitoring and evaluation of poverty alleviation projects along the dictates of the MDGs. CWIQ was conducted using the National Integrated Survey of Households (NISH) design run by the NBS. A representative sample of urban and rural was selected in each of the 36 States and Federal Capital Territory (FCT). A total of 7,740 Enumeration Areas (EAs) were selected with an estimated 77,400 housing units (HU) nationwide. The education information in the surveys are accessibility to schools, educational attainment, adult literacy, primary school and secondary school enrolment, types of school attended(private or public), scholarship award, school drop out and interruption, satisfaction with school, reasons for school dropout and interruption, education expenditure(tuition fees, cost of book, boarding fees, cost of transportation to schools.



Figure 1: Map of Nigeria.

6.2 Data analysis

Descriptive statistics and econometric analyses were applied to relevant data on education from Nigeria. Descriptive statistics such as mean and percentage distribution were employed in the analysis. In econometric analysis, I drew from the framework of Behrman and Knowles (1999) in relating income with schooling. Their framework can be stated as

$$S = f(Y) - - - (1)$$

Where S is schooling indicator and Y is mean household income. In their study they used age when starting school, grade passed per year of school, last completed grade and exam score in the last completed grade as measure of schooling indicator. However, these indicators of schooling can be biased by child innate ability. I then used accessibility to primary and secondary school as schooling indicator in the model. As Table 1 has shown different scholars have used various measure of school indicator, but many of these indicators can be biased by child innate ability as pointed out by Behrman and Knowles (1999). Moreover, in order to avoid omitted variable bias, I introduced regional dummy variable into equation (1) to capture regional disparity in accessibility to school in Nigeria, which researchers assume can affect accessibility to education in Nigeria (Aluede, 2006). This regional dummy can also capture the effect of differences in educational policies between the regions. Hence, equation (1) is modified as

$$S = f(Y, R) - - (2)$$

Where S is accessibility to school, measured as number of minutes it takes a student to reach the nearest school. The benchmark is 25 minutes³. Student that can get to school within this period is said to have access to school, the proportion those students in percentage is taken as school accessibility, the proportion of those that spend more than 25 minutes gives the idea of students without access(CWIQ, 2006)⁴. Y is household mean income measured in Naira, R is region dummy, where southern states are given 1, and zero otherwise. Equation 2 is then estimated for primary and secondary schools.

In estimating the relationship between income distribution and accessibility to schools, i employed Psacharopoulos (1977) approach. He stated his econometric model as

$$GINI=f(EDINEQ, Yp) - - - (3)$$

_

³ The mean time of getting to nearest primary and secondary school in Nigeria is 25 minutes (CWIQ, 2006)

⁴ It should be noted that there are different measure of accessibility by different scholar as long as it represents the interest of the study. Sackey (2007) used distance that take 30 minutes to student house as measure of accessibility to education in Ghana. Students that spend less than 30 minutes to get to their schools are said to have access. Mainardi (2007) used also physical distance to health facility as measure of accessibility to health care.

Where GINI is Gini coefficient describing income distribution, EDINEQ is educational inequality, as measured by the coefficient of variation of enrolment by school level. In modifying this approach I used Gini for household income for each state. I then constructed accessibility to education for each state. In order to avoid the problem of multicollinearity⁵, I have replaced mean income in equation 3 with regional dummy, to have

$$GINI = f(ACCESS, R) - - - (4)$$

While GINI is the Gini coefficient, describing income distribution in each state in Nigeria, ACCESS is the mean of percentage of students with access to school in each state in Nigeria. R is regional dummy as defined previously.

In order to examine the effect of change in mean income on income distribution which Psacharopoulos (1977) was able to estimate with his equation 3, I then estimated a separate equation that relates mean household income with income distribution. This is stated as

$$GINI = f(Yp, R) - - -(5)$$

Where GINI and R are as defined previously. Yp is mean disposable income for each state.

In deciding the equation of best fit, I tried Linear, Power, Semi-log and Exponential functional forms. Econometric, economic and statistical criteria were used to select the equation of best fit.

clusion of these highly positively correlated variables may lead to problem of multicollinearity. However, the correlation between accessibility to school and regional dummy is low, being 0.153 and -0.310 for primary and secondary school respectively. The fact that there is regional disparity in income distribution in Nigeria is a popular knowledge (Oyekale et al., 2006). All these may justify the inclusion of regional dummy in equation 4.

⁵ The correlation between mean income and accessibility to school is greater than 0.80, they are in fact 0.911 and 0.960 for primary and secondary school respectively. Therefore, the in-

7 RESULTS AND DISCUSSIONS OF DESCRIPTIVE ANALYSIS

7.1 Accessibility to Primary and secondary Education in Nigeria on Regional Basis

The result of analysis of the data indicates that 76% of children in Nigeria had access to primary school education. South-West zone recorded the highest figure of 88.0 per cent, followed by North-Central, 79.7 per cent, while the South-East recorded the lowest rate (60.6 per cent) as presented in Table 3. However, about 47 per cent of Nigerian children had access to secondary school education. The South-West zone recorded the highest figure of 69.4 per cent, followed by South-South with 48.0 per cent access, while the South-East recorded the lowest rate with 32.3 per cent. The fact that there are more accessibility to primary and secondary education in South-West zone in Nigeria can be predicated on the fact that the spread of education start from South-West Zone through missionary (Imahe and Alabi, 2005). Aluede (2006) traced this disparity in education in Nigeria to the history of educational development in the country. He said that the history of educational development in an environment can as well explain the cause of educational disparity in that particular environment. He demonstrated that Western form of education came into Nigeria through the activities of the missionaries. This greatly determined which area was to have early taste of western education and the areas that were not to experience early missionary activities. Thus, Kosemani (1993) observed that the British had set up the colony of Sierra Leone in 1784 as a haven for freed slaves. Later many New World bound slaves who were rescued by British Squadrons were resettled in Sierra Leone. As it turned out, many of those captured were Nigerians mostly Yorubas from the Southwest zone of Nigeria. By 1870, many of them have been converted to Christianity and have become wealthy enough to plan a return to their father land. The effect of this movement back home on Christianity and western education in Nigeria was tremendous. As slave settlers in Sierra Leone, they had received religious, academic and some technical education and a taste of European ways of life. This accident of history brought along with the returnee Nigerians Christianity. This in turn led the early ex-slave settlers in Abeokuta (South West zone) and other areas in West Africa to request for Christian evangelists to come over and 'civilize' the indigenous Africans. The arrival of the missionaries brought with it western education and civilization.

Moreover, in Nigeria, politics brought about disparity in educational development. The political activities of the 1950s to 1960 brought about changes in educational development. The Macpherson Constitution of 1951 sowed the seed of disparity in terms of educational development. It among other things, gave powers to the regional governments to pass laws on education. The result of this was the presentation of a comprehensive proposal for the introduction of free universal and compulsory primary education other wise known as universal Primary Education (UPE) to the Western (South West) House of Assembly by chief Awolowo. The implementation of the programme took effect in the Western Region in January, 1955. Since then the government in South-western Nigeria gives more priority to education than any other zones in the country. The government in South West Nigeria implemented Free Education Policy between 1979 and 1984 and since then the incoming government has been subsiding education in one form or the other (Aluede, 2006).

Other causes of educational disparities apart from history in some countries that experienced colonization are the fact that early development of schooling in specific areas was due to proximity to the coast, to the existence of climatic conditions conducive to missionary settlement and or to the presence of particular local rulers or forms of political authority that were hospitable (for whatever reasons) to the spread of western type of schools (Foster, 1980). The important thing is that the coastal towns, that had European settlements, had early contact with western education and civilization while others who settled in the interior could not have early contact with neither the missionaries nor the colonial masters. This may explain high accessibility to education in coastal South Western Nigeria. The fact is that early centres of educational development have proved to maintain their initial advantages over relatively long periods. That perhaps may explain why Abeokuta and Ondo areas in Yoruba land (South West) tend to maintain their educational lead in Nigeria. This initial disparity in level of educational development has continued to expand since more advanced regions are able to capitalise on their earlier educational traditions (Aluede, 2006). Various authors have isolated ethnicity, economic, cultural factors as factors that are also responsible for disparity in distribution of educational resources in Nigeria (Kosemani, 1992; 1996; Okwonko, 1988)

The lowest accessibility to education in South East poses a challenge to the government of South East zone in increasing accessibility to basic education in the zone. Although there is general opinion that many of the children in the zone take to

business and commerce instead of going to school, it is possible that they are not going to school because they don't have access to them. Alabi (2008) has demonstrated that the relationship between accessibility to basic education and school enrolment is positive, especially for secondary school education in Nigeria, suggesting that the enrolment in schools by children in South East in Nigeria can be lower when compared with the other zones. The lower accessibility to education in South Eastern Nigeria is also an indication of lower priority given by the succeeding government in South eastern zone. This is because the educational opportunities in the South east zone were closed to that of South West before independence (Aluede, 2006).

The consequence of disparity in accessibility to education in Nigeria is a big challenge to balanced growth and development. The economic implication of this disparity is that it has been observed to be bringing about inequality in economic, political and social development as well. It has also been observed that this disparity in educational development can cause disintegration in nations in the course of struggles for positions and power. Alabi and Abu (2008) have attributed pronounced crisis in Niger-Delta to low educational status of the indigenes of the region. Deng (2003) also reported that the war in Southern Sudan can be associated with the way the education opportunities are distributed in the country.

Further analysis indicates that the accessibility to and the satisfaction with basic education in Nigeria are factors that are positively correlated (see table 3). About 58 per cent of the children of primary school age (6 to 11 years old)⁶ expressed satisfaction with primary education service. Majority of the children in the South-West (76.8 per cent) were satisfied with their education, while a little over half the number of children in the North-West (58.0 per cent), South-South (54.8 per cent) and North-Central (50.3 per cent) were satisfied. The least satisfaction rate was recorded in the North-East (37.2 per cent). More than half (56.6 per cent) the number of children in secondary schools expressed satisfaction with their secondary education. The level of satisfaction was highest in the South-West (74.8 per cent), followed by North-West (58.2 per cent), while the least satisfaction rate was in the North-East (42.1 per cent). Available data on education in Nigeria indicated that satisfaction with education and enrolment rate are positively related. The correlation coefficients are 0.63 and 0.45 for

_

⁶ More than 22 million children are 6 to 11 years old in Nigeria, the official primary school age in Nigeria (Huebler, 2005). The official secondary school age in Nigeria is 12 to 17 years.

primary and secondary education (CWIQ, 2006). This implies that satisfaction increases with school enrolment. The trend of satisfaction will affect the attendance in school. Table 3 reveals that the national net enrolment in primary and secondary schools in Nigeria are 61.5% and 45.6% respectively. This is not significantly different from the estimate of Huebler (2005). He reported that only 60.1% of all children of primary school age were attending primary school in Nigeria. He indicated that boys had a higher net attendance rate (NAR) than girls, with 63.7% compared to 56.5% for girls in primary school. For secondary school he estimated that 35.1% of the children were in secondary school. His disaggregated result indicated that the boys' secondary school net attendance rate (NAR) was 37.5% and for girls it was 32.6%. The UNICEF also indicated low enrolment in Nigerian schools. It reported in 2008 that more than 10 million Nigerian children are out of school (Punch, 2008). Out of the 10 million, 4.7 million and 5.3 million are of primary and secondary school age respectively. The report says further that sixty-two per cent of the children out of school are girls. This is one of the reasons for UNICEF to make Nigeria one of the priority countries for girls' education (Huebler, 2005).

Table 3: Accessibility to Primary and Secondary Schools on Regional Basis in Nigeria (%)

Schooling Indicators	North East	North West	North Central	South East	South West	South South	National
Accessibility to Primary School	71.9	76.4	79.7	60.6	88.0	71.7	75.9
Accessibility to Secondary	36.3	44	47.7	32.3	69.4	48.0	47.3
Satisfaction with Primary School	37.2	58.0	50.3	65.1	76.8	54.8	58.3
Satisfaction with Secondary School	42.1	58.2	48.5	53.4	74.8	49.1	56.6
Net Enrolment Primary School	43.7	42.2	72.5	81.6	82.3	76.8	61.5
Net Enrolment Secondary School	25.8	25.4	46.1	59.7	64.9	58.7	45.6

Source: Computed from CWIQ, 2006

7.2 Accessibility to Basic Education in Nigeria on Location Basis

The growth of industries brought about urban and rural settlements in Nigeria. The companies attracted educated men and women as workers in the industries. These workers on their parts struggled to ensure that their children received good education. This brought about a situation where schools in urban areas were well staffed, equipped and financed while rural schools experienced poor staffing, furnishing and financing. In some cases, teachers refused transfer to rural areas (Aluede, 1998). This created gap in schooling in rural and urban areas in Nigeria. The result of analysis of accessibility to primary and secondary education in rural and urban areas in Nigeria indicates that accessibility to primary school education in urban areas is 86.7 per cent, while the rural areas recorded 71.9 per cent (see table 4). The urban-rural gap to secondary education in Nigeria is wider, about 32% gap. The accessibility to secondary education in urban areas is 69.3 per cent, while the accessibility in rural areas is 37.5 per cent. The disparity in satisfaction with education in rural and urban follows the same trend with disparity in satisfaction with primary and secondary education in Nigeria. In the rural areas, 54.5 per cent of primary school children were satisfied with their schools, when compared with 66.7 per cent in the urban areas. Among the students that were satisfied with their secondary schools, 51.6 per cent in the rural are satisfied while 64.9 per cent in the urban areas are satisfied with their secondary schools. Table 4 indicates that the gap between enrolment in primary school in rural and urban areas in Nigeria is 18%, while the gap between enrolment in secondary school in rural and urban areas is about 20% all in favour of urban areas.

This disparity in education system in rural and urban areas has been noted by other scholars. Huebler (2005) also noted that children in urban areas had a higher primary NAR (69.5%) than children in rural areas (55.7%) in Nigeria. The secondary school NAR in urban areas was 46.3% and in rural areas it was 28.7%. A study to determine whether educational discrepancies exist between urban and rural sections of Liberia revealed that only 40% or fewer of rural schools followed the prescribed curricula. They demonstrated that level of teacher qualification was low, a greater problem in rural schools, half of whose teachers had a high school education or less (Coleman and Clark, 1983). Other major findings were that all school facilities were below standards required for effective school programs; a general lack of funding for education existed, and ministry reports verified an imbalance in the central educa-

tional administration, to the detriment of rural schools, primarily because of problems of transportation and communication. Hazans and Trapeznikora (2008) reported that rural location can be an obstacle to accessibility to secondary school education. They showed that the distance between the student and the nearest school location on the average was 2.41km in urban area and that it was 5.87km in rural area. They also indicated that the percentage of secondary school teachers with higher education in rural area was lower than in urban area.

However, the disparity in schooling based on location is contrary to the tenet of philosophy of education in Nigeria. According to the Nigerian Education Policy Document Section 1(4), subsection c 'Every Nigerian child shall have right to equal educational opportunities irrespective of any real or imagined disabilities each according to his or her ability' (National Policy on Education, 2004: 7). The Policy states that 'The philosophy of education therefore is based on the provision of equal access to educational opportunities for citizens of the country at the primary, secondary and tertiary levels both inside and outside the formal school system'. Section 1(7), subsection e also state that 'Universal Basic Education in a variety of forms, depending on needs and possibilities, shall be provided for all citizens' (National Policy on Education, 2004: 9).

Table 4: Accessibility to Basic education in Urban and Rural Areas in Nigeria (%)

	Urban	Rural	Difference
Accessibility to Primary School	86.7	71.9	14.8
Accessibility to Secondary School	69.3	37.5	31.8
Satisfaction with Primary school	66.7	54.5	12.2
Satisfaction with Secondary school	64.9	51.6	13.3
Enrolment in Primary school	74.6	56.6	18.0
Enrolment in secondary school	59.3	39.6	19.7

Source: Computed from CWIQ, 2006

7.3 Accessibility to Basic Education and Income Distribution in Nigeria

Table 5 indicates that the disparity in school enrolment between children from the richest and poorest households is great. In the richest 20% of all households, 82.9% of all children of primary school age attended primary school. In the poorest

20% of all households only two out of five children were in school (primary NAR 40.4%). The table also shows that 63.8% of children from the richest 20% of all households were in secondary school, compared to only 14.6% of children from the poorest 20% of all households. The differences between the accessibility to primary and secondary schools in richest and poorest households are about 43% and 49% respectively in favour of the richest households as indicated in Table 5. This suggests that the school attendance rate is strongly linked to household wealth. The fact that children from the poor households are left out of educational opportunities may pose a great danger to poverty reduction and income inequality in Nigeria. This suggests that that income distribution in favour of rich households will lead to reduction in school enrolment in Nigeria. Since educational inequality is positively related to earning (Chiswick, 1971), the educational inequality in Nigeria may prevent the children of the poor to have access to income earning opportunities that will take them out of poverty.

Some other studies have also shown that there are wealth differences in school enrollment and attainment in most developing countries, but the gaps vary widely across countries. It is argued that that the differences between rich and poor are particularly large (more than 45 percentage points) in several West African countries –Benin, Burkina Faso, Mali, and Senegal. In contrast, small differences are seen in Kenya and Malawi. Why are enrollment rates lower and educational outcomes worse among the poor? Because it is harder for poor children to reach school: the latter tend to be concentrated in cities and areas where the wealthier households reside (UNECA, 2003). For example, in Guinea, the average travel time to the nearest primary school is 47 minutes in rural areas but only 19 minutes in urban areas (Ministère de l'éducation pré-universitaire et de l'éducation civique, 2001). It is also observed that in many African countries, the widespread use of child labour (particularly in rural areas) very often interferes with children's attendance at school. Hence, in these areas, school enrolment and drop-out rates are much worse. In multi-ethnic countries, the drop-out rates among ethnic minorities is also higher than that of dominant groups. Evidence also shows that in many countries, children from the low classes lag behind in educational achievement (UNECA, 2003). There is a high likelihood that access to public goods is skewed towards higher income brackets. For example children not completing primary education in Mali and Morocco are over 30% for the lowest 40% income group, while the top 20% income bracket had a much lower drop out rate, approximately 10% for Mali and less than 5% for Morocco (UNECA, 2003).

Table 5: Accessibility to primary and secondary education and income distribution in Nigeria

	Richest 20%	Poorest 20%	Difference
Access to Primary School	85.1	42.4	42.7
Access to Secondary	66.2	17.0	49.2
School			
Enrolment in Primary	82.9	40.1	42.8
school			
Enrolment in secondary	63.8	14.6	49.2
school			

Sources: Computed from CWIQ, 2006 and Huebler (2005)

Table 6 goes further to establish the fact that the children from the poor in rural and urban areas have different access to educational opportunities in favour of the poor in urban areas. The Table shows that accessibility to primary education among the urban and rural poor are about 68% and 41% respectively, while accessibility to secondary school among the urban and rural poor are about 45% and 29% respectively. If it is understood that most of the farmers are located in rural areas in Nigeria, the fact that there will continuing cycle of poverty among these poor who are feeding Nigerian teeming population cannot be far-fetched. This confirms that accessibility to basic education is compounded when you are poor and located in the rural area in Nigeria.

In accounting for low educational level of rural dwellers in Nigeria, Abidogun (2008) indicated that many teachers in Nigeria reject posting into the rural areas while those that do not reject the posting treat their presence in such areas as a 'part time assignment'. Yet, effectively educating the rural population that make up over 60% of the country is a necessary precondition for national development. Education is one of the key strategies for rural development since it propels the development of desirable attitudes that are favourable to change and for technical progress. Anyaegbu et al. (2004) thus opined that rural education is the key to rural development and an essential building block of national development; that poverty cannot be eradicated without eliminating illiteracy among the rural populace and without finding a systematic way to raise their level of knowledge. According to Abidogun (2008: 4) there is the 'general consensus that the rate of agricultural development and rural transformation is

directly related to the educational standard of the rural communities'. Such rural education programme will widen rural populace's horizon and predispose them to greater receptivity of new ideas.

Table 6: Accessibility to primary and secondary education among the poor in rural and urban areas in Nigeria

	Urban Poor	Rural Poor
Access to Primary School	68.4	41.2
Access to Secondary School	44.8	29.4
Satisfaction with Primary school	58.2	44.7
Satisfaction with Secondary school	55.9	39.0
Enrolment in Primary school	64.3	49.9
Enrolment in secondary school	44.8	29.4

Source: Computed from CWIQ, 2006

8 RESULTS AND DISCUSSIONS OF ECONOMETRIC ANALYSES

8.1 The Effect of Household Income and Accessibility to Basic Education

In assessing the effect of income on accessibility to education in Nigeria, exponential functional form is selected as lead equation and presented in Table 7. Table 6 indicates that mean household income and regional dummy can explain 73% influence of income on accessibility to primary and secondary education in Nigeria. The F-values are significant at 10%. The regional dummy is not significant, indicating that the difference in accessibility to primary and secondary education in Nigeria on regional basis is not significant. Mean household income is a positive and significant determinant of accessibility to basic education in Nigeria. The coefficients of mean income for primary and secondary school equations are 0.92 and 0.88 respectively, suggesting that increase in mean household income will increase accessibility to primary school more significantly than secondary school. The positive effect of income and accessibility to education estimated in Table 7 has also been reported by Tansel

(1977) and Montgomery and Kouame (1993) for Cote d d'Ivoire, Deolalikar (1997) for Kenya, and Birdsall and Orivel (1996) for Mali. However, the marginal effects of 0.92 and 0.88 estimated in this study are greater than the average of about 0.20 elasticity estimated for other African countries. The differences in my estimates and other African countries can be predicated on the differences in measure of schooling. This study is based on the accessibility to school, whereas other studies were based on school attendance, student teacher ratio, school attainment etc as indicated in Table 1. Moreover, Behrman and Knowles (1999) pointed out that income elasticity estimate of schooling depends on the mean income of a country.

It is interesting to note that income and regional dummy explain only 73% variation in accessibility to basic education and that constant coefficient is significant in the equation. The economic implication of this is that there are other variables that influence accessibility to basic education that are not included in this model. The case of rural and urban disparity has been discussed in this context on Table 4. The other effect can be from policy perspective. Alabi (2008) has demonstrated that favourable education policy in 1990 resulted in building more primary schools than in 2006 in Nigeria. He also shows that while Nigeria has 116101 classrooms in secondary schools in 2000, this was reduced to 98734 in 2006, due to the fact that the government did not build new classroom and nor repaired the collapsed ones. The concomitant effects of these may be low accessibility and low enrolment in basic education in Nigeria.

Table 7: Effect of Household Income on Accessibility to Basic Education in Nigeria

	Primary	School	Secondary	School
Variable	Coefficient	t-ratio	Coefficient	t-ratio
Constant	337.53	3.23*	502.70	3.35*
Household Income	0.92	3.92*	0.88	3.68*
Region	0.35	1.46	0.128	0.54
F	7.90**		7.67**	
Adjusted R Square	0.73		0.73	

Dependent Variable: Accessibility to Primary and Secondary School (%)

(Exponential Functional Form) * Significant at 5%** Significant at 10%

Source: Author Estimates

8.2 Effect of Accessibility to Basic Education on Income Distribution on in Nigeria

Table 8 shows that accessibility to basic school and regional dummy can explain 99% variation in income inequality in Nigeria. The F-values indicates that the equations are significant at 5%. The regional dummy is negative and significant at 5%. This suggests that income redistribution in favour of Northern region will reduce income inequality in Nigeria. Accessibility to primary and secondary school coefficients are negative and significant, implying that increase in access to primary and secondary schools in Nigeria will reduce income inequality in Nigeria. This finding is in consonance with report of Psacharopolous (1977) that a policy aiming at equalization of access to different level of education might help in reducing income inequality. Becker and Chiswick (1966) also demonstrate (in the US) that income inequality is positively correlated with schooling inequality. Chiswick (1971) study shows that earnings inequality increases with educational inequality. The Chenery and Syrquin (1975) finding supports the fact that higher level of schooling reduces income inequality.

Since accessibility coefficient for secondary school (-1.04) is greater than that of primary school (-1.00), it implies that income inequality in Nigeria can be equalized faster by increasing access to secondary school than by increasing access to primary school. This may be due to the fact that the return to secondary school education in Nigeria is greater that of primary school (Okuwa, 2004).

Table 8: Effect of Accessibility to Basic Education on Income Inequality in Nigeria.

	Primary	School	Secondary	School
Variable	Coefficient	t-ratio	Coefficient	t-ratio
Constant	0.88	109.29*	.72	95.40*
Accessibility	-1.00	-39.91*	-1.04	-21.99*
Region	-0.31	-12.35*	-0.17	3.51*
F	816.39*		247.84*	
Adjusted R Square	0.99		0.99	

Dependent Variable: Income Inequality (GINI Coefficient)

(Linear Functional Form) * Significant at 5%

Source: Author Estimates

8.3 Effect of Mean Household Income on Income Distribution

In Table 9 I estimated the relationship between mean income and income inequality in Nigeria. This done to demonstrate that accessibility to basic education has greater influence in reducing income inequality in Nigeria than mean household income. Table 8 indicates that regional dummy and mean income can explain 85% change in income inequality in Nigeria. The F-value suggests that the equation is significant at 5%. The regional dummy is not significant. The mean income is significant and negatively related to income inequality. This indicates that increase in household income in Nigeria will reduce income inequality in Nigeria. However, this increase has to be in favour of low income households. The economic implication is that economic growth in Nigeria that will lead to increase in disposable income is still desirable in Nigeria. More importantly, when the coefficient of accessibility to basic education (-1.00 for primary and -1.04 for secondary) in Table 7 is compared with the coefficient of mean income in Table 9, the coefficients of accessibility to basic education are greater than the coefficient of mean income. This is in conformity with finding of Psacharopolous (1977). He indicated that while the coefficient of education inequality was 0.18, that of mean income was 0.11. This implies that policy that increase accessibility to basic education will redistribute income faster than policy that increase mean household income

Table 9: Effect of Mean Household Income on Income Inequality in Nigeria

Variable	Coefficient	t-ratio
Constant	0.73	22.14*
Mean Household Income	-0.98	-5.33*
Region	-0.125	-0.68
F	14.59*	
Adjusted R Square	0.85	

Dependent Variable: Income Inequality (GINI Coefficient)

(Linear Functional Form)* Significant at 5%

9 CONCLUSIONS AND RECOMMENDATIONS

This study establishes that there is unequal access to basic education between the poor and non poor in Nigeria. Household income is an important determinant of access to basic education in Nigeria. Increase in access to basic education can redistribute income in Nigeria faster than increase in household income. The income redistribution effect of accessibility to secondary school is greater than primary school. This study concludes that a policy aiming at equalisation of access to primary and secondary school education might help in reducing income inequality in Nigeria.

I can therefore recommend pro-poor growth in Nigeria that will lead to increase income in favour of the poor. This is type of growth that will be beneficial to the poor segment in the society. Poverty alleviation strategies that target mainly the poor can be in this category. The economic intervention agencies should encourage the formation of cooperative societies, through which poor can pull their funds together to increase their scale of their income generation operations.

Policy that will increase access to basic education school should be implemented in Nigeria. This can be in form of free basic education, given of scholarship to the brilliant but indigent students, school subsidy etc. There is need to build more schools and classrooms and rehabilitate the existing ones⁷. There should be a policy and programmes to build more schools in rural areas and closer to where people live, his where the assistance of federal government will be highly needed. Since, the federal government collects more than 50% of revenue in Nigeria; it has to assist the state and local government in building primary and secondary schools. Notwithstanding, as pointed out (World Bank, 2004), even though expanding access is important it is not a sufficient condition to ensure that all children from different backgrounds are enrolled and progress is made in the education system. This means that apart from physical expansion of school infrastructure, context specific policy measures are required to create effective demand for education among poor households and individuals by federal, state and local government.

School buses can be provided at subsidized price for the student by NGOs, Private organisations and philanthropic initiatives. A situation where about 88% and

their schools in the past five years (CWIQ, 2006).

-

⁷ About 71% of students in basic schools claim that there was no new building construction in their school in the past five years, while and 61% of the students claim that there was no rehabilitation in

65% of students of primary and secondary schools respectively trek to their school, for average of about 25 minutes in Nigeria⁸ (CWIQ, 2006), may not be conducive for optimum academic performance and retention. A virile Parent Teacher Association (PTA) association can be of help in the provision of some of there basic education resources.

There is also need to stem up allocation to education sector in Nigeria. The present 0.76% of GNP allocated to education sector may not be enough to increase accessibility to education in Nigeria. Moreover, this is far lower than average of about 4.5% of GNP allocated to education sector in Sub-Sahara African countries (Abidogun, 2008)⁹, and lower still when compared with average of 6 % of GDP allocated to education sector by OECD countries(Robert et al, 2002). The annual budgetary allocation of about 10% to education sector between 1995 and 2008 is too low when compared with 26%, 21% and 21% in Ghana, Botswana and Kenya respectively (Dike, 2008). In fact, UNESCO recommended 26% budgetary allocation to education sector in order to increase accessibility and enrolment in schools (Abidogun, 2008).

Finally, in promoting basic education in Nigeria, the tuition fees should be moderated to be affordable by the people. Books can be supplied free through Education Trust Fund of the Federal Government (ETF), United Nations agencies, NGO and other donor agencies. The Parent Teachers' Association (PTA), Rotary Club, Lion Clubs and other philanthropic organizations can assist in reducing the cost education in the state by providing scholarship, books and vehicle to the needy schools and students. UNICEF and UNESCO can come to aid of Nigeria in funding school building and rehabilitation. They should also assist in carrying out an updated survey of Nigerian education system. The most recent survey on education in Nigeria was conducted in 2006; this is even part of poverty and welfare survey. A separate survey on education system is urgently needed in Nigeria.

⁸ The available data indicates that about 9% and 26% of students in Nigeria primary and secondary school respectively spend more than one hour to get to their schools

⁹ Countries such as South Africa, Kenya and Cote d Ivoire allocate 7.9%, 6.5% and 5.0% of their GNP respectively to their education sector (Dike, 2008). It is also far below the 15% recommended by UNICEF (Abidogun, 2008).

REFERENCES

- Abebaw, D, Delelegn, A. and A. Assefa (2007), Determinants of Child Schooling Progress in Rural Ethiopia, Available on the internet at http://www.gdnet.org/middle.php?oid=237&zone=docs&action=doc&doc=11814
- Abidogun, B. G (2008), Education Sector Reforms and Childhood Education for Rural

 Development in Nigeria, Available on the internet at www.transformedu.org/LinkClick.

 aspx?fileticket=C5TtZ%2FfGt2Y%3D&tabid=71&mid=416 -
- Alabi, Reuben Adeolu (2008), Comparative analysis of socio economic constraints in Niger-Delta, Nigeria. European Journal of Economics, Finance and Administrative sciences 10: 63-74.
- Alabi, R. A. and S.O. Abu (2008), Analysis of Health, Education and Environmental Conditions In Coastal Area Of Nigeria: The Case Of Niger-Delta, In Proceedings of 1st Postgraduate Researcher's Conference on Meeting the Environmental Challenges in the Coastal Region of Nigeria. University of Abertay, Dundee, Scotland, United Kingdom, 29-30th September, 2008.
- Aluede, R. O. A. (2006), Regional Demands and Contemporary Educational Disparities in Nigeria, J. Soc. Sci., 13(3): 183-189.
- Aluede, R. O. A. (1998), A Comparative Analysis of Some Compensatory Education Schemes in Australia, Canada, Nigeria and the United States of America. Unpublished doctoral dissertation, Port Harcourt: University of Port Harcourt.
- Anyaegbu R., Christman, D. E. and C. Jingpu (2004), The Challenges of Rural Education in Nigeria: Potential Lessons from China. E-Journal of Teaching and Learning In Diverse Settings, I (2) 160-174.
- Becker, Gary S. (1967), Human Capital and the Personal Distribution of Income: An Analytical Approach, Woytinsky Lecture, Ann Arbor: University of Michigan Press, Republished in Gary S. Becker, 1975, Human Capital, Second Edition. New York: National Bureau of Economic Research.
- Bedi, A. S., Kimalu, P. K., Manda, D.K., and Nafula, N. (2004), The Decline in Primary Enrollment in Kenya, Journal of African Economies, Vol.13, No. 1., pp. 1-43.

- Behrman, Jere R., and James C. Knowles (1999), Household Income and Child Schooling in Vietnam, The World Bank Economic Review Vol. 13(2): 211-256.
- Behrman, Jere R., and James C. Knowles (1997), How Strongly Is Child Schooling Associated with Household Income?, Department of Economics, University of Pennsylvania, Philadelphia.
- Behrman, Jere R., and Paul Taubman (1986), Birth Order, Schooling, and Earnings, Journal of Labor Economics 4(4):S121-45.
- Behrman, Jere R., Nancy Birdsall, and Robert Kaplan (1996), "The Quality of Schooling and Labor Market Outcomes in Brazil: Some Further Explorations." In (eds.), Nancy Birdsall and Richard Sabot, *Opportunity Forgone: Education in Brazil*. Baltimore, Md. Johns Hopkins University Press.
- Birdsall, Nancy, and François Orivel (1996), Demand for Primary Schooling in Rural Mali: Should User Fees Be Increased?, In: Education Economics 4(3):279-96.
- Burnhill, P., Garner, C., and A. McPherson (1990), Parental Education, Social Class and Entry to Higher Education 1976-86, Journal of the Royal Statistical Association, Series A, 153(2), 233-248.
- Burke, K. and Beegle, K. (2004), Why Children Aren't Attending School: The Case of Northern Tanzania, in: Journal of African Economies, 13(2), 333-355.
- Chenery, H., and M. Syrquin (1975), Patterns of Development, 1950–970, Oxford, London.
- Chiswick, B. (1971), Earnings Inequality and Economic Development, in: Quarterly Journal of Economics, LXXV pp. 21–39
- Cockburn, J. and Dostie, B. (2007), Child Work and Schooling: The Role of Household Asset Profiles and Poverty in Rural Ethiopia. Journal of African Economies, February, 1-45.
- Cochrane, Susan Hill, Kaplana Mehra, and Ibrahim Taha Osheba (1986), The Educational Participation of Egyptian Children. Education and Training Series Report EDT45. Washington, D.C. World Bank.
- Cogneau, D and S. Mesple Somps(2008), Inequality of Opportunity for Income in Five Countries of Africa. Available on the internet at http://www.dial.prd.fr/dial_publications/PDF/Doc_travail/2008-04.pdf

- Coleman, A. B and E.J Clark (1983), An analysis of the rural-urban balance for education in developing countries: A case study of Liberia. Available on the internet at http://eric.ed.gov/ERICWebPortal/custom/portlets/recordDetails/detailmini.jsp?_nfpb=true&_&ERICExtSearch_SearchValue_0=ED234976&ERICExtSearch_SearchType_0=no&accno=ED234976
- Collier, P., (2000), Economic Causes of Civil Conflict and their Implications for Policy, mimeo, Washington, D.C.: World Bank.
- CWIQ (2006), Core Welfare Indicator Questionnaire Survey Results, National Bureau of Statistics, Abuja, Nigeria
- Deng, L. B. (2003), Education in Southern Sudan: War, Status and challenges of achieving Education For All Goals, UNESCO EFA Monitoring Report. Available on the internet at portal.unesco.org/education/en/files/25755/10739022851Education in Southern Sudan. War, status_and_challe.
- Deolalikar, Anil B. (1997), The Determinants of Primary School Enrolment and Household Expenditures in Kenya: Do They Vary by Income?, Department of Economics, University of Washington, Seattle.
- Dercon, S. Selassie, T.G. and P.Krishan (2005), 'The Urban Labour Market during Structural Adjustment in Ethiopia', In: (eds.) Bigsten, A, Kebede, B. and A. Shimeles, Poverty, Income Distribution and Labour Markets in Ethiopia, Nordiska Africainstitutet, Sweden. Also available on the internet at http://www.nai.uu.se/publications/download.html/91-7106-526-1.pdf?id=25095
- Dike, V. (2008), The State of Education in Nigeria and the Health of the Nation, Available on the internet at http://www.africaeconomicanalysis.org/articles/14/1/The-state-of-education-in-Nigeria-and-the-health-of-the-nation/Page1.html
- EFA Global Monitoring Report (2002), Education for All: Is the World on Track, Paris: UNESCO
- Fischer, R. D. and P. J. Serra, (1996), Income convergence within and between countries, in: International Economic Review 37(3), 531-551.
- Foster, P. (1980), Regional disparities in educational development: some critical observations", in: Regional Disparities in Educational Development: A Controversial Issue, G. Carron and T. N. Chau (eds.). Paris: UNESCO: International Institute for Educational Planning.

- Frankema, E and Jutta Bolt (2006), Measuring and Analysing Educational Inequality: The Distribution of Grade Enrolment Rates in Latin America and Sub-Saharan Africa. Available on the internet at http://www.ggdc.net/pub/gd86.pdf
- Friedman, Milton (1962), Capitalism and Freedom, Princeton, N.J.: Princeton University Press.
- Gitter, S. R. and Barham, B. L (2007), Credit, Natural Disasters, Coffee, and Educational Attainment in Rural Honduras, in: World Development, Vol. 35, No. 3, pp. 498-511.
- Glaeser, E.L. (1994), Why does schooling generate economic growth?, In: Economics Letters 44 (3), 333-337.
- Glewwe, Paul, and Hanan Jacoby (1994), Student Achievement and Schooling Choice in Low-Income Countries: Evidence from Ghana. Journal of Human Resources 29(3):842-64.
- Glewwe, Paul, and Hanan Jacoby (1995), An Economic Analysis of Delayed Primary School Enrollment and Childhood Malnutrition in a Low-Income Country, in: Review of Economics and Statistics 77(1):156-69.
- Glomm, G and Ravikumar, B (1992), Public versus private investment in human capital: Endogenous growth and Income Inequality, in:. Journal of Political Economy 100: 818-834.
- Gordon, A. and E. Catherine (2002), Rural Non Farm Activities And Poverty Alleviation In Sub-Saharan Africa, Natural Resources Institute Policy Series 14, Also available on internet at: www.nri.org/publications/policyseries/PolicySeries
- Gundlach, E.J.N. de Pablo and N. Weiset (2001),. Education is good for the Poor: A note on Dollar and Kraay, WIDER Discussion Paper No. 2001/137.
- Gunther, R (2002), (Re) distribution of personal income, education and economic performance across countries, WIDER Discussion PaperNo.2002/34, Pp1-34
- Hannum, E. and Buchmann, C (2005), Global Educational Expansion and Socio-Economic Development: An Assessment of Findings from the Social Sciences. World Development, 33(3), 333-354.
- Hanushek, E.A (1986). The economics of schooling: production and efficiency in public schools, in: Journal of Economic Literature, XXIV, 1141-1177.
- Hanushek, E.A. (2002), Publicly provided education, in: Auerbach, A., Feldstein, M.

- (eds), Handbook of Public Economics (Amsterdam: North Holland).
- Hazans, M and Ija Trapeznikova (2008), Access to Secondary Education in Albania: Incentives, Obstacles, and Policy Spillovers, Available on the internet at www.gdnet.org/pdf2/gdn_library/annual_conferences/Ninth_annual_conference/Trapeznikova.ppt.
- Huebler, F. (2005), International Education Statistics, Available on the internet at http://huebler.blogspot.com/2005/10/primary-school-attendance-in nigeria.html
- Igbuzor, O (2008), The State Of Education In Nigeria. Available on the internet at http://www.dawodu.com/igbuzor14.htm
- Imahe, O.J and R. A. Alabi (2005), Government Educational Spending and Human Capita Development in Nigeria: Empirical Evidence from Nigeria. Paper Presented at 'Regional Conference on Education in West Africa.: Constraints and Opportunities' Dakar, Senegal, Organized by Cornell University, CREA, and INRA, Paris. 1st--2nd November, 2005
- Islam, N. (1997), The non-farm sector and rural development: Review of issues and evidence, Food, Agriculture and the Environment Discussion Paper 22, International Food Policy Research Institute, Washington D.C.
- Jacoby, Hanan (1994), Borrowing Constraints and Progress through School: Evidence from Peru, in: Review of Economics and Statistics 76(1):151-60.
- Knight, J. and Sabot R. (1983), Educational expansion and the Kuznets effect, in: American Economic Review 73 (5): 1132-1136
- Kosemani, J.M. (1992), The ethnic factor in educational disparity in Nigeria, Bensu Journal of Education, 3(1): 15-25.
- Kosemani, J.M. (1993), The historical background to contemporary educational disparity in Nigeria", in: Kosemani, J.M (ed.), Educational Disparity: The Nigeria Situation. Lagos: Africa Heritage Research and Publications.
- Kosemani, J.M. (1996), "The political context of educational inequality in Nigeria", A Collection of Kosemani's Publications (1), Port Harcourt: University of Port Harcourt.
- Lavy, Victor (1996), School Supply Constraints and Children's Educational Outcomes in Rural Ghana, in: Journal of Development Economics 51(2):291-314.

- Lee, J. W. and R. J. Barro (2001), Schooling quality in a cross-section of countries, in: Economica 68, 465-488.
- Lire, E (2005), Child Labor and Schooling Decisions in Urban and Rural Areas: Comparative Evidence from Nepal, Peru, and Zimbabwe, in: World Development, 33(3), 455-480.
- Lloyd, C. B. and Hewett, P. C. (2004), Universal primary schooling in Sub-Saharan Africa: is gender equity enough?, Paper presented at the international conference 'Social Policies and Human Rights for Children and Women: Achieving and Monitoring the Millennium Development Goals', April 28-30, 2004.
- Luca, R. (1988), On mechanics of economic development, in: Journal of Monetary Economics, 22: 3-42
- Mainardi, S. (2007), Unequal Access to Public HealthCare Facilities: Theories and Measurement Revisited, *Surveys in Mathematics and its Application*, Vol. 2: 91-112.
- Ministère de l'enseignement pré-universitaire et de l'éducation civique, Service statistiques et planification Guinea (2001), Schooling in Guinea: Findings from the GDHS-II 1999, Calverton, Maryland.
- Montgomery, Mark R., and Aka Kouame (1993), Fertility and Child Schooling in Cote d'Ivoire: Is There a Tradeoff?, Department of Economics, State University of New York at Stony Brook.
- NBS (2006), National Bureau of Statistics, Abuja, Nigeria, Also available on the internet at http://www.nigerianstat.gov.ng/
- National Policy on Education (2004), Federal Republic of Nigeria National Policy on Education, 4th edition, Nigerian Educational Research and Development Council, Yaba, Lagos, Nigeria.
- Okojie C. E. E. (2002), Gender and Education as Determinants of Household Poverty in Nigeria, World Institute for Development Economics Research Discussion Paper, No. 2002/37
- Okuwa, O.B. (2004), Private returns to Higher education in Nigeria, AERC RP 139, African Economic Research Consortium, Nairobi, Kenya.
- Okonkwo, C. E. (1988), Introduction to Comparative Education, Owerri: Totan Publishers Limited.
- Psacharopoulos, George (1994), Returns to Investment in Education: A Global Update, in: World Development 22(9):1325-44.

- Psacharopoulos, George (1977), Unequal Access to Education and Income Distribution, in: De Economist 125(3): 383-392. Also available on the internet at www.springerlink.com/content/m451p82q3r45835w
- Punch (2008), 10 million Children out of School in Nigeria UNICEF, Available on the internet at http://www.punchng.com/Articl.aspx?theartic=Art200801160352 620
- Rosenzweig, Mark R. (1995), Why Are There Returns in Schooling?, in: American Economic Review 85(2):153-58.
- Rosenzweig, Mark R., and Kenneth J. Wolpin (1986), Evaluating the Effects of Optimally Distributed Public Programs, in: American Economic Review 76 (3): 470-87.
- Sahn, D.E. and Stifel, D. C. (2004), Urban-rural inequality in living standards in Africa, *WIDER Research Paper* No. 2004/4, United Nations University.
- Subrahmanian, R. (2002), Citizenship and the Right to Education, Institute for Development Studies, Bulletin Vol. 33 No. 2
- Ram, R (1989), Can Education Expansion Reduce Income Inequality in Less-Developed Countries?, in: Economics of Education Review 8:186-190.
- Robert, A. Dur, J. and C.N. Teulings (2002), Education, Income distribution and Public Policy, Available on the internet at http://cep.lse.ac.uk/seminarpapers/29-11-02-TEU.pdf
- Sackey, H. A (2007), Private Returns to education in Ghana: Implications for Investments in Schooling and Migration, African Economic Research Consortium (AERC) Research Paper 174. Nairobi, Kenya
- Schultz, T.P. (1999), Health and schooling investments in Africa, in: Journal of Economic Perspectives 12 (3): 67-88.
- Tamura, R. (1991), Income convergence in an endogenous growth model, in: Journal of Political Economy 99: 522-526.
- Tansel, Aysit (1997), School Attainment, Parental Education, and Gender in Cote d'Ivoire and Ghana, in: Economic Development and Cultural Change 45(4): 825-56.
- World Bank (1996), Vietnam Education Financing Sector Study: A Sector Report .Report 15925-VN, Washington, D.C.

UNECA

- (2003),Pro poor growth Strategies in Africa, http://www.uneca.org/eca_resources/mee
 tings_events/espd/poorgrowth/supply_side%20propoor%20education.final.i.e.p
 df
- UNESCO (1999), Education for All, Available on the internet at athttp://www.unesco.org/education/efa/wef_2000/regional_frameworks/frame_africa.shtml
- United Nations (2000), Millennium Development Goals, www.development goals.org
- Van de Walle, Dominique, and Kimberly Nead, eds. (1995), Public Spending and the Poor: Theory and Evidence. Baltimore, Md.: Johns Hopkins University Press for the World Bank.
- World Bank (2004), World Development Report 2004: Making Services Work for the Poor People. New York: Oxford University Press.
- Zilcha, I. and Viaene, Jean-Marie (2003), Public and private provision of education and income inequality, Discussion Paper No.7-2003, The Pinhas Sapir Centre for Development, Tel-Aviv University, http://econ.tau.ac.il/sapir.

Bisher erschienene "Berichte aus dem Weltwirtschaftlichen Colloquium" des Instituts für Weltwirtschaft und Internationales Management

(Downloads: http://www.iwim.uni-bremen.de/publikationen/pub-blue)

Nr. 1 Sell, Axel:

Staatliche Regulierung und Arbeitslosigkeit im internationalen Sektor. 1984. 35 S.

Nr. 2 Menzel, Ulrich/ Senghaas, Dieter:

Indikatoren zur Bestimmung von Schwellenländern. Ein Vorschlag zur Operationalisierung. 1984. 40 S.

Nr. 3 Lörcher, Siegfried:

Wirtschaftsplanung in Japan. 1985. 19 S.

Nr. 4 Iwersen, Albrecht:

Grundelemente der Rohstoffwirtschaftlichen Zusammenarbeit im RGW. 1985. 52 S.

Nr. 5 Sell. Axel.

Economic Structure and Development of Burma. 1985. 39 S.

Nr. 6 Hansohm, Dirk/ Wohlmuth, Karl:

Transnationale Konzerne der Dritten Welt und der Entwicklungsprozeß unterentwickelter Länder. 1985. 38 S.

Nr. 7 Sell, Axel:

Arbeitslosigkeit in Industrieländern als Folge struktureller Verhärtungen. 1986. 21 S.

Nr. 8 Hurni, Bettina:

EFTA, Entwicklungsländer und die neue GATT-Runde. 1986. 28 S.

Nr. 9 Wagner, Joachim:

Unternehmensstrategien im Strukturwandel und Entwicklung der internationalen Wettbewerbsfähigkeit. 1986. 28 S.

Nr. 10 Lemper, Alfons:

Exportmarkt Westeuropa. Chinas Vorstoß auf die Weltmärkte. 1987. 40 S.

Nr. 11 Timm, Hans-Jürgen:

Der HWWA-Index der Rohstoffpreise - Methodik, Wirtschafts- und Entwicklungspolitische Bedeutung. 1987. 57 S.

Nr. 12 Shams. Rasul:

Interessengruppen und entwicklungspolitische Entscheidungen. 1987. 23 S.

Nr. 13 Sell, Axel:

ASEAN im Welthandelskraftfeld zwischen USA, Japan und EG. 1987. 23 S.

Nr. 14 Kim, Young-Yoon/Lemper Alfons:

Der Pazifikraum: Ein integrierter Wirtschaftsraum? 1987. 24 S.

Nr. 15 Sell, Axel:

Feasibility Studien für Investitionsprojekte, Problemstruktur und EDV-gestützte Planungsansätze. 1988. 18 S.

Nr. 16 Hansohm, Dirk/ Wohlmuth, Karl:

Sudan's Small Industry Development. Structures, Failures and Perspectives. 1989. 38 S.

Nr. 17 Borrmann, Axel/ Wolff, Hans-Ulrich:

Probleme bei der Planung industrieller Investitionen in Entwicklungsländern. 1989. 28 S.

Nr. 18 Wohlmuth, Karl:

Structural Adjustment and East-West-South Economic Cooperation: Key Issues. 1989. 53 S.

Nr. 19 Brandtner, Torsten:

Die Regionalpolitik in Spanien unter besonderer Berücksichtigung der neuen Verfassung von 1978 und des Beitritts in die Europäische Gemeinschaft. 1989. 40 S.

Nr. 20 Lemper, Alfons:

Integrationen als gruppendynamische Prozesse. Ein Beitrag zur Neuorientierung der Integrationstheorie. 1990. 47 S.

Nr. 21 Wohlmuth, Karl:

Die Transformation der osteuropäischen Länder in die Marktwirtschaft - Marktentwicklung und Kooperationschancen. 1991. 23 S.

Nr. 22 Sell. Axel:

Internationale Unternehmenskooperationen. 1991. 12 S.

Nr. 23 Bass, Hans-Heinrich/Li, Zhu:

Regionalwirtschafts- und Sektorpolitik in der VR China: Ergebnisse und Perspektiven. 1992. 28 S.

Nr. 24 Wittkowsky, Andreas:

Zur Transformation der ehemaligen Sowjetunion: Alternativen zu Schocktherapie und Verschuldung. 1992. 30 S.

Nr. 25 Lemper, Alfons:

Politische und wirtschaftliche Perspektiven eines neuen Europas als Partner im internationalen Handel. 1992. 17 S.

Nr. 26 Feldmeier, Gerhard:

Die ordnungspolitische Dimension der Europäischen Integration. 1992. 23 S.

Nr. 27 Feldmeier, Gerhard:

Ordnungspolitische Aspekte der Europäischen Wirtschafts- und Währungsunion. 1992. 26 S.

Nr. 28 Sell, Axel:

Einzel- und gesamtwirtschaftliche Bewertung von Energieprojekten. - Zur Rolle von Wirtschaftlichkeitsrechnung, Cost-Benefit Analyse und Multikriterienverfahren-. 1992. 20 S.

Nr. 29 Wohlmuth, Karl:

Die Revitalisierung des osteuropäischen Wirtschaftsraumes - Chancen für Europa und Deutschland nach der Vereinigung. 1993. 36 S.

Nr. 30 Feldmeier, Gerhard:

Die Rolle der staatlichen Wirtschaftsplanung und -programmierung in der Europäischen Gemeinschaft. 1993. 26 S.

Nr. 31 Wohlmuth, Karl:

Wirtschaftsreform in der Diktatur? Zur Wirtschaftspolitik des Bashir-Regimes im Sudan. 1993. 34 S.

Nr. 32 Shams, Rasul:

Zwanzig Jahre Erfahrung mit flexiblen Wechselkursen. 1994. 8 S.

Nr. 33 Lemper, Alfons:

Globalisierung des Wettbewerbs und Spielräume für eine nationale Wirtschaftspolitik. 1994. 20 S.

Nr. 34 Knapman, Bruce:

The Growth of Pacific Island Economies in the Late Twentieth Century. 1995. 34 S.

Nr. 35 Gößl, Manfred M./ Vogl. Reiner J.:

Die Maastrichter Konvergenzkriterien: EU-Ländertest unter besonderer Berücksichtigung der Interpretationsoptionen. 1995. 29 S.

Nr. 36 Feldmeier, Gerhard:

Wege zum ganzheitlichen Unternehmensdenken: "Humanware" als integrativer Ansatz der Unternehmensführung. 1995. 22 S.

Nr. 37 Gößl. Manfred M.:

Quo vadis, EU? Die Zukunftsperspektiven der europäischen Integration. 1995. 20 S.

Nr. 38 Feldmeier. Gerhard/ Winkler. Karin:

Budgetdisziplin per Markt oder Dekret? Pro und Contra einer institutionellen Festschreibung bindender restriktiver Haushaltsregeln in einer Europäischen Wirtschafts- und Währungsunion. 1996. 28 S.

Nr. 39 Feldmeier, Gerhard/ Winkler, Karin:

Industriepolitik à la MITI - ein ordnungspolitisches Vorbild für Europa? 1996. 25 S.

Nr. 40 Wohlmuth, Karl:

Employment and Labour Policies in South Africa. 1996. 35 S.

Nr. 41 Bögenhold, Jens:

Das Bankenwesen der Republik Belarus. 1996. 39 S.

Nr. 42 Popov, Djordje:

Die Integration der Bundesrepublik Jugoslawien in die Weltwirtschaft nach Aufhebung der Sanktionen des Sicherheitsrates der Vereinten Nationen. 1996. 34 S.

Nr. 43 Arora, Daynand:

International Competitiveness of Financial Institutions: A Case Study of Japanese Banks in Europe. 1996. 55 S.

Nr. 44 Lippold, Marcus:

South Korean Business Giants: Organizing Foreign Technology for Economic Development. 1996. 46 S.

Nr. 45 Messner, Frank:

Approaching Sustainable Development in Mineral Exporting Economies: The Case of Zambia. 1996. 41 S.

Nr. 46 Frick, Heinrich:

Die Macht der Banken in der Diskussion. 1996. 19 S.

Nr. 47 Shams, Rasul:

Theorie optimaler Währungsgebiete und räumliche Konzentrations- und Lokalisationsprozesse. 1997. 21 S.

Nr. 48 Scharmer, Marco:

Europäische Währungsunion und regionaler Finanzausgleich - Ein politisch verdrängtes Problem. 1997. 45 S.

Nr. 49 Meyer, Ralf/ Vogl, Reiner J.:

Der "Tourismusstandort Deutschland" im globalen Wettbewerb. 1997. 17 S.

Nr. 50 Hoormann, Andreas/Lange-Stichtenoth, Thomas:

Methoden der Unternehmensbewertung im Akquisitionsprozeß - eine empirische Analyse -. 1997. 25 S.

Nr. 51 Gößl, Manfred M.:

Geoökonomische Megatrends und Weltwirtschaftsordnung. 1997. 20 S.

Nr. 52 Knapman, Bruce/ Quiggin, John:

The Australian Economy in the Twentieth Century. 1997. 34 S.

Nr. 53 Hauschild, Ralf J./ Mansch, Andreas:

Erfahrungen aus der Bestandsaufnahme einer Auswahl von Outsourcingfällen für Logistik-Leistungen. 1997. 34 S.

Nr. 54 Sell, Axel:

Nationale Wirtschaftspolitik und Regionalpolitik im Zeichen der Globalisierung - ein Beitrag zur Standortdebatte in Bremen. 1997. 29 S.

Nr. 55 Sell. Axel:

Inflation: does it matter in project appraisal. 1998. 25 S.

Nr. 56 Mtatifikolo, Fidelis:

The Content and Challenges of Reform Programmes in Africa - The Case Study of Tanzania, 1998. 37 S.

Nr. 57 Popov, Djordje:

Auslandsinvestitionen in der BR Jugoslawien. 1998. 32 S.

Nr. 58 Lemper, Alfons:

Predöhl und Schumpeter: Ihre Bedeutung für die Erklärung der Entwicklung und der Handelsstruktur Asiens. 1998. 19 S.

Nr. 59 Wohlmuth, Karl:

Good Governance and Economic Development. New Foundations for Growth in Africa. 1998. 90 S.

Nr. 60 Oni, Bankole:

The Nigerian University Today and the Challenges of the Twenty First Century. 1999. 36 S.

Nr. 61 Wohlmuth, Karl:

Die Hoffnung auf anhaltendes Wachstum in Afrika. 1999. 28 S.

Nr. 62 Shams, Rasul:

Entwicklungsblockaden: Neuere theoretische Ansätze im Überblick. 1999. 20 S.

Nr. 63 Wohlmuth, Karl:

Global Competition and Asian Economic Development. Some Neo-Schumpeterian Approaches and their Relevance. 1999. 69 S.

Nr. 64 Oni, Bankole:

A Framework for Technological Capacity Building in Nigeria: Lessons from Developed Countries. 1999. 56 S.

Nr. 65 Toshihiko, Hozumi:

Schumpeters Theorien in Japan: Rezeptionsgeschichte und gegenwärtige Bedeutung. 1999. 22 S.

Nr. 66 Bass, Hans H.:

Japans Nationales Innovationssystem: Leistungsfähigkeit und Perspektiven. 1999. 24 S.

Nr. 67 Sell, Axel:

Innovationen und weltwirtschaftliche Dynamik – Der Beitrag der Innovationsforschung nach Schumpeter. 2000. 31 S.

Nr. 68 Pawlowska, Beata:

The Polish Tax Reform. 2000. 41 S.

Nr. 69 Gutowski, Achim:

PR China and India – Development after the Asian Economic Crisis in a 21st Century Global Economy. 2001. 56 S.

Nr. 70 Jha, Praveen:

A note on India's post-independence economic development and some comments on the associated development discourse. 2001. 22 S.

Nr. 71 Wohlmuth, Karl:

Africa's Growth Prospects in the Era of Globalisation: The Optimists versus The Pessimists. 2001. 71 S.

Nr. 72 Sell. Axel:

Foreign Direct Investment, Strategic Alliances and the International Competitiveness of Nations. With Special Reference on Japan and Germany. 2001. 23 S.

Nr. 73 Arndt, Andreas:

Der innereuropäische Linienluftverkehr - Stylized Facts und ordnungspolitischer Rahmen. 2001. 44 S.

Nr. 74 Heimann, Beata:

Tax Incentives for Foreign Direct Investment in the Tax Systems of Poland, The Netherlands, Belgium and France. 2001. 53 S.

Nr. 75 Wohlmuth, Karl:

Impacts of the Asian Crisis on Developing Economies – The Need for Institutional Innovations, 2001, 63 S.

Nr. 76 Heimann, Beata:

The Recent Trends in Personal Income Taxation in Poland and in the UK. Crisis on Developing Economies – The Need for Institutional Innovations. 2001. 77 S.

Nr. 77 Arndt, Andreas:

Zur Qualität von Luftverkehrsstatistiken für das innereuropäische Luftverkehrsgebiet. 2002. 36 S.

Nr. 78 Frempong, Godfred:

Telecommunication Reforms - Ghana's Experience. 2002. 39 S.

Nr. 79 Kifle, Temesgen:

Educational Gender Gap in Eritrea. 2002. 54 S.

Nr. 80 Knedlik, Tobias/Burger, Philippe:

Optimale Geldpolitik in kleinen offenen Volkswirtschaften – Ein Modell. 2003. 20 S.

Nr. 81 Wohlmuth, Karl:

Chancen der Globalisierung - für wen? 2003. 65 S.

Nr. 82 Meyn, Mareike:

Das Freihandelsabkommen zwischen Südafrika und der EU und seine Implikationen für die Länder der Southern African Customs Union (SACU). 2003. 34 S.

Nr. 83 Sell. Axel:

Transnationale Unternehmen in Ländern niedrigen und mittleren Einkommens. 2003. 13 S.

Nr. 84 Kifle, Temesgen:

Policy Directions and Program Needs for Achieving Food Security in Eritrea. 2003. 27 S.

Nr. 85 Gutowski, Achim:

Standortqualitäten und ausländische Direktinvestitionen in der VR China und Indien. 2003. 29 S.

Nr. 86 Uzor, Osmund Osinachi:

Small and Medium Enterprises Cluster Development in South-Eastern Region of Nigeria. 2004. 35 S.

Nr. 87 Knedlik, Tobias:

Der IWF und Währungskrisen - Vom Krisenmanagement zur Prävention. 2004. 40 S.

Nr. 88 Riese, Juliane:

Wie können Investitionen in Afrika durch nationale, regionale und internationale Abkommen gefördert werden? 2004. 67 S.

Nr. 89 Meyn, Mareike:

The Export Performance of the South African Automotive Industry. New Stimuli by the EU-South Africa Free Trade Agreement? 2004. 61 S.

Nr. 90 Kifle, Temesgen:

Can Border Demarcation Help Eritrea to Reverse the General Slowdown in Economic Growth? 2004. 44 S.

Nr. 91 Wohlmuth, Karl:

The African Growth Tragedy: Comments and an Agenda for Action. 2004. 56 S.

Nr. 92 Decker, Christian/ Paesler, Stephan:

Financing of Pay-on-Production-Models. 2004. 15 S.

Nr. 93 Knorr, Andreas/Žigová, Silvia:

Competitive Advantage Through Innovative Pricing Strategies – The Case of the Airline Industry. 2004. 21 S.

Nr. 94 Sell, Axel:

Die Genesis von Corporate Governance. 2004. 18 S.

Nr. 95 Yun, Chunji:

Japanese Multinational Corporations in East Asia: Status Quo or Sign of Changes? 2005.

57 S.

Nr. 96 Knedlik, Tobias:

Schätzung der Monetären Bedingungen in Südafrika. 2005. 20 S.

Nr. 97 Burger, Philippe:

The transformation process in South Africa: What does existing data tell us? 2005. 18 S.

Nr. 98 Burger, Philippe:

Fiscal sustainability: the origin, development and nature of an ongoing 200-year old debate. 2005. 32 S.

Nr. 99 Burmistrova, Marina A.:

Corporate Governance and Corporate Finance: A Cross-Country Analysis. 2005. 16 S.

Nr. 100 Sell, Axel:

Alternativen einer nationalstaatlichen Beschäftigungspolitik. 2005. 41 S.

Nr. 101 Bass, Hans-Heinrich:

KMU in der deutschen Volkswirtschaft: Vergangenheit, Gegenwart, Zukunft. 2006. 19 S.

Nr. 102 Knedlik, Tobias/ Kronthaler, Franz:

Kann ökonomische Freiheit durch Entwicklungshilfe forciert werden? Eine empirische Analyse. 2006. 18 S.

Nr. 103 Lueg, Barbara:

Emissionshandel als eines der flexiblen Instrumente des Kyoto-Protokolls. Wirkungsweisen und praktische Ausgestaltung. 2007. 32 S.

Nr. 104 Burger, Phillipe:

South Africa economic policy: Are we moving towards a welfare state? 2007. 31 S.

Nr. 105 Ebenthal, Sebastian:

Messung von Globalisierung in Entwicklungsländern: Zur Analyse der Globalisierung mit Globalisierungsindizes. 2007. 36 S.

Nr. 106 Wohlmuth, Karl:

Reconstruction of Economic Governance after Conflict in Resource-Rich African Countries: Concepts, Dimensions and Policy Interventions. 2007. 42 S.

Nr. 107 Smirnych, Larissa:

Arbeitsmarkt in Russland - Institutionelle Entwicklung und ökonomische Tendenzen. 2007. 34 S.

Nr. 108 Kifle, Temesgen:

Africa Hit Hardest by Global Warming Despite its Low Greenhouse Gas Emissions. 2008. 22 S.

Nr. 109 Alabi, Reuben Adeolu:

Income Sources Diversification: Empirical Evidence from Edo State, Nigeria. 2008. 54 S.

Nr. 110: Jantzen, Katharina

Eine Einführung in Regulierungssysteme für die Fischerei im Nordatlantik am Beispiel der Fanggründe vor Island und Neufundland, 2008, 32 S.

Nr. 111: Ebenthal, Sebastian:

Mexiko im Kontext der Globalisierung - Ergebnisse eines Globalisierungsindex für Entwicklungsländer, 2008, 72 S.

Nr. 112: Rieckmann, Johannes

Two Dynamic Export Sectors (Diamonds, Tourism) in Namibia and Botswana: Comparison of Development Strategies . 2008, 48 S

Nr. 113 Albers, Harm:

Globalisierung und Wachstum: Die Konvergenzdebatte – dargestellt mit Bezug auf Veröffentlichungen dreier Wissenschaftler. 2008. 35 S.

Nr. 114 **Reuben Adeolu Alabi:** Income Distribution And Accessibility To Primary And Secondary Schools In Nigeria. 2008. 49 S.